Author's response to reviews

Title: Effects of epidural lidocaine analgesia on labor and delivery: A randomized, prospective, controlled trial

Authors:
Shahram Nafisi - (nafisish@yahoo.com)

Version: 6 Date: 16 December 2006

Author's response to reviews: see over
Dear The BioMed Central Editorial Production Team

Thanks from your email.

Here is the point-by-point description of the changes.

1- The author name in the manuscript is forename then surname.
2- The author's qualification was removed.
3- Unnecessary capitals were removed.
4- The trial registration number mentioned in the abstract.
5- The job title removed from the affiliation.
6- Information regarding funding provided in the Acknowledgements section.
7- Keywords section removed.
8- Introduction - renamed to 'Background.'
9- The type of bracket used is consistent and the reference is before the full stop.
10- Following the Conclusions, a section titled 'Competing interests' is added.
11- There are author contribution and acknowledgements sections.
12- Roman numerals deleted.
13- Blank pages were deleted.

Regards,

Shahram Nafisi